– 218 –

6.2 Conference Contributions

XXX. International Winter Meeting on Nuclear Physics Bormio, Italy, January 27 - February 1, 1992

H. Daniel Nuclear excitation by antiproton annihilation at rest Proceedings, ed. I. Iori, Univ. di Milano, Suppl. 91, 1992

C. Fuchs, L. Sehn, H.H. Wolter Relativistic Kinetic Calculations of Heavy Ion Collisions with Phase Space Dependent Fields Proceedings, p. 540, ed. I. Iori, Univ. of Milano, Suppl. 91, 1992

A. Gillitzer, J. Homolka, F. Schumacher, E. Berdermann, P. Kienle, W. Koenig, A. Schröter, H. Ströher, B. Povh Subthreshold Antiproton Production in Heavy Ion Collisions at SIS-Energies Proceedings, p. 392

H.-J. Körner Produktion of Neutron-Rich Nuclei in Heavy Ion Collision Proceedings, p. 515

A. Weiß, P. Egelhof, B. Franzke, H. Geissel, A. Gruber, W. Henning, G. Kraus, A. Magel, G. Münzenberg, F. Nickel, M. Hamm, J.V. Kratz, J. Friese, A. Gillitzer, H.-J. Körner, M. Peter, J.P. Schiffer, L.V. Chulkov, M. Golovkov, A.A. Ogloblin Nuclear Structure Studies Far Off Stability Using Radioactive Beams and Inverse Kinematics Proceedings7

Physics of High Energy Density in Matter Hirschegg, Kleinwalsertal, Austria, February 03 - 07, 1992

A. Ulrich Lasers pumped by heavy-ion beams Proceedings

– 219 –

Workshop on Ion Beam Applications HMI Berlin, February 17 - 18, 1992

W. Assmann, G. Dollinger Materials analysis with HI at the Munich MP-Tandem

6th European Symposium on Semiconductor Detectors Milano, Italy, February 24 - 26, 1992

T. Beha, W. Dünnweber, P. Klein Radiation damage in detector-amplifier structures capable of signal storage Proceedings, p. 386

Frühjahrstagung Kernphysik der DPG Salzburg, Austria, February 24 -28, 1992

B. Ackermann, H. Baltzer, T. Bihn, C. Ensel, V. Grafen, C. Günther, J. Manns, M. Marten-Tölle, U. Müller, R. Tölle, J. de Boer, N.A. Gollwitzer Collective E1 Transitions in Even-Even Actinide Nuclei Verhandl. DPG (VI) 27 (1992) 212

P. Ackerbauer, W.H. Breunlich, M. Fuchs, S. Fussy, M. Jeitler, P. Kammel, B. Lauss, J. Marton, W. Prymas, J. Werner, J. Zmeskal, K. Lou, C. Petitjean, P. Baumann, H.Daniel, F.J. Hartmann, W. Schott, T. von Egidy, P. Wojciechowski, D. Chatellard, J.-P. Egger, E. Jeannet, T. Case, K.M. Crowe, R.H. Sherman, V. Markushin Muonic hydrogen and deuterium x-ray detection in muon catalyzed fusion with CCDS Verhandl. DPG (VI) 27 (1992) 31

K. Balog, K.E.G. Löbner, Th. Winkelmann A Time Recording and Data Acquisition-System for Time-of-Flight Measurements in the ESR Verhandl. DPG (VI) 27 (1992) 151

B. Bante, U. Bittner, T. Faestermann, A. Gillitzer, G. Korschinek, D. Müller Detection of^{90}Sr with Accelerator Mass Spectrometry Verhandl. DPG (VI) 27 (1992) 154

P. Baumann, H. Daniel, F.J. Hartmann, M. Mühlbauer, W. Schott, P. Wojciechowski, A. Fuchs, K. Lou, C. Petitjean, D. Taqqu, P. Hauser, F. Kottmann Measurement of the stopping power for μ^− and μ^+ at energies between 3 and 90 keV Verhandl. DPG (VI) 27 (1992) 32

– 220 –

P. Baumann, H. Daniel, T. von Egidy, S. Grunewald, F.J. Hartmann, R. Lipowsky, E. Moser, W. Schott, P. Ackerbauer, W.H. Breunlich, M. Fuchs, M. Jeitler, P. Kammel, J. Marton, N. Nägele, J. Werner, J. Zmeskal, V. Markushin, H. Bossy, K.M. Crowe, R.H. Sherman, C. Petitjean, K. Lou Results on muon catalyzed pt fusion Verhandl. DPG (VI) 27 (1992) 30

P. Baumann, H. Daniel, F.J. Hartmann, M. Mühlbauer, W. Schott, P. Wojciechowski Calculations for a device for cooling charged particle beams by moderation plus acceleration Verhandl. DPG (VI) 27 (1992) 124

E. Berdermann, P. Kienle, W. Koenig, A. Schröter, A. Gillitzer, J. Homolka, F. Schumacher, H. Ströher, B. Povh Subthreshold Antiproton Production in Heavy Ion Collisions at SIS-Energies Verhandl. DPG (VI) 27 (1992) 177

W. Borchert, M. Haft, H. Lenske, H.H. Wolter Relativistic direct reactions with heavy ions Verhandl. DPG (VI) 27 (1992) 53

A. Cazan and the Tritron-Team The Munich Postaccelerator Tritron Verhandl. DPG (VI) 27 (1992) 143

L.V. Chulkov, P. Egelhof, J. Friese, H. Geissel, A. Gillitzer, M. Golovkov, M. Hamm, W. Henning, H.-J. Körner, J.V. Kratz, G. Kraus, A. Magel, G. Münzenberg, F. Nickel, A.A. Ogloblin, M. Peter, J.P. Schiffer, K. Sümmerer, D.J. Vieira, A. Weiß Nuclear Structure Studies using Radioactive Beams and Inverse Kinematics Verhandl. DPG (VI) 27 (1992) 122

M. Ciric, J. Friese, H.-J. Körner, M.R. Maier A Low Cost Analog Pulse Processing System for Use with Silicon PIN Diodes Verhandl. DPG (VI) 27 (1992) 152

J. Cub, M. Bussas, K.-H. Speidel, W. Karle, U. Knopp, H. Busch, H.-J. Wollersheim, K. Vetter, J. Gerber, C. Ender, F. Köck, F. Hagelberg Transient magnetic field in Gd for^{32}S(2^+) ions at the 1s electron Bohr velocity Verhandl. DPG (VI) 27 (1992) 39

J. Cub, K.-H. Speidel, U. Knopp, H. Busch, S. Kremeyer, H.-J. Wollersheim, N. Martin, X. Hong, K. Vetter, N.A. Gollwitzer, A. Booten, K. Heyde Transient field measurements in Gd for^{40}Ar(2^+)-ions at the 1s electron Bohr velocity and the g-factor of the 2^+-state at 1.461 MeV Verhandl. DPG (VI) 27 (1992) 129

– 221 –

H. Daniel, K.-H. Hiddemann and O. Schwentker Neutrino masses from the tritium beta spectrum Verhandl. DPG (VI) 27 (1992) 88

H. Daniel, F.J. Hartmann, W. Neumann, W. Schott Measurement of the stopping power of Au and MgF_2 for slow muons Verhandl. DPG (VI) 27 (1992) 124

R. Fauser, H.H. Wolter Correlation functions in quantum field theories for non-equilibrium systems Verhandl. DPG (VI) 27 (1992) 54

J. Friese, A. Gillitzer, H.-J. Körner, J. Reinhold, M.R. Maier The SISWICH as a Detector Telescope with Intrinsic Calibration Verhandl. DPG (VI) 27 (1992) 140

J. Friese, H.-J. Körner, J. Reinhold, R. Schneider, H. Trieb, K. Zeitelhack, B. Blank, T. Brohm, H. Geissel, G. Münzenberg, Y. Fujita, F. Nickel, M. Pfützner, K. Rykaczewski, I. Schall, D. Schardt, A. Schröter, M. Steiner, K. Sümmerer, B. Voss, J. Weckenmann Production of Neutron-rich Nuclei by Fragmentation of 760 A·MeV Xe-projectiles Verhandl. DPG (VI) 27 (1992) 156

C. Fuchs, L. Sehn, H.H. Wolter Covariant kinetic calculations of heavy ion collisions using phase space dependent fields Verhandl. DPG (VI) 27 (1992) 66

N.A. Gollwitzer, J. de Boer, A. Lösch, A.I. Levon, K. Kaiser, F. Rieß, R.G. Stokstad, E.G. Vogt, F. Wagenpfeil, M. Wieland Cross-Sections for the one-neutron transfer to^{199}Pt Verhandl. DPG (VI) 27 (1992) 205

F. Heine, T. Faestermann, A. Gillitzer Proton and alpha radioactivity in the vicinity of^{100}Sn Verhandl. DPG (VI) 27 (1992) 131

T. Henkelmann, G. Belajev, V. Dubenkov, A. Golubev, G. Korschinek, S. Latyshev, B. Sharkov, A. Shumshurov, B. Wolf Charge state distribution of tantalum ions produced simultaneously by CO_2- and Nd:YAG laser from a laser ion source Verhandl. DPG (VI) 27 (1992) 190

M. Hennes, H. Lenske, H.H. Wolter Statistical properties of realistic nuclear models Verhandl. DPG (VI) 27 (1992) 116

– 222 –

R.R. Hilton, W. Höhenberger, H.J. Mang, and P. Ring Systematics of K=1^+ States in Rare Earth Nuclei Verhandl. DPG (VI) 27 (1992)

B. Hinfurtner, E. Hagn, C. König, E. Zech NICOLE und ISOLDE Kollaboration Kernresonanzmessungen an orientierten Y, Zr und Nb Isotopen Verhandl. DPG (VI) 27 (1992) 80

P. Jänker, K. Balog, H. Leitz, U. Lenz, K.E.G. Löbner, M. Moralles, H.-G. Thies, Th. Winkelmann A Stringent Investigation of Compound Nuclei Decay Verhandl. DPG (VI) 27 (1992) 224

M. Jeitler, W.H. Breunlich, M. Cargnelli, P. Kammel, J. Marton, P. Pawlek, J. Werner, J. Zmeskal, K.M. Crowe, J. Kurck, A. Janett, C. Petitjean, R.H. Sherman, H. Bossy, H. Daniel, W. Neumann Epithermal effects in muon catalyzed dt fusi Verhandl. DPG (VI) 27 (1992) 32

U. Knopp, K.-H. Speidel, J. Cub, W. Karle, H. Busch, S. Kremeyer, J. Gerber, F. Hagelberg The magnetic moment of the^{54}Fe(2^+_1 ) state from transient field measurements Verhandl. DPG (VI) 27 (1992) 40

C. König, E. Hagn, B. Hinfurtner, E. Zech NICOLE and ISOLDE collaboration Messungen von Kernmomenten neutronenarmer Ta und Ir Isotope Verhandl. DPG (VI) 27 (1992) 134

S. Kremeyer, H. Busch, K.-H. Speidel, J. Cub, W. Karle, U. Knopp, M. Bussas, J. Gerber, P. Maier- Komor A precision plunger at low temperature for detection of polarized ions on emergence from ferromagnetic Gd Verhandl. DPG (VI) 27 (1992) 155

H. Lenske Structure of nuclei at the neutron drip line Verhandl. DPG (VI) 27 (1992) 205

M.S. Lotfranaei, T. Haninger, Y.S. Kim, P. Hofmann, F.J. Hartmann, T. von Egidy, H. Märten, A. Ruben The opening angle of fragments from^{236}U and^{252}Cf fission Verhandl. DPG (VI) 27 (1992) 94

A. Lösch, J. de Boer, N.A. Gollwitzer, A.I. Levon, G. Graw, R. Hertenberger, H. Kader, V. Grafen, B. Ackermann, T. Bihn, C. Günther, R.S. Simon Spectroscopy of^{229}Pa Verhandl. DPG (VI) 27 (1992) 211

– 223 –

C. Nunes, M. Schaden, and P. Ring Heavy Mesons in a Gluonic Vacuum Verhandl. DPG (VI) 27 (1992)

M. Reboiro and P. Ring Models for the Nucleus as a Condensate of Quark Clusters Verhandl. DPG (VI) 27 (1992)

I. Romanski, E. Hagn, P. Herzog, B. Hinfurtner, J. Prinz, E. Zech NICOLE and ISOLDE collaboration On-line Nuclear Orientation of Neutron deficient Au-Pt-Ir Nuclei Verhandl. DPG (VI) 27 (1992) 193

W. Schmid, P. Baumann, H. Daniel, T. von Egidy, F.J. Hartmann, P. Hofmann, Y.S. Kim, M.S. Lotfranaei, D. Hilscher, D. Polster, H. Rossner, J. Jastrzebski, A. Grabowska, W. Kurcewicz, P. Lubinski, A. Stolarz Measurement of antiproton induced fission Verhandl. DPG (VI) 27 (1992) 94

H. Schmidt, J. Friese, H.-J. Körner, G. Beier, R. Ecker, Wa. Mayer, L. Müller, W. Wagner, G. Rosner Investigation of^{186}W Induced Heavy Ion Reactions at Incident Energies Near the Coulomb Barrier Verhandl. DPG (VI) 27 (1992) 206

R. Siudak, H. Lenske, H.H. Wolter Inelastic heavy ion scattering in a multistep direct reaction description Verhandl. DPG (VI) 27 (1992) 206

M. Siemaszko, W. Dünnweber, W. Hering, M. Herrlein, D. Konnerth, S. Kovats, D. Steierwald Statistical model description of γ-ray decay of the highly excited compound nucleus^{32}S Verhandl. DPG (VI) 27 (1992) 117

A. Surowiec, W. Dünnweber, W. Hering, D. Konnerth, W. Zipper, J. Gerl, D. Habs, H.J. Hennrich, V. Metag, D. Schwalm, R.S. Simon Resonances of the m = 0 and 2 substates in^{12}C +^{12}C →^{12}C +^{12}C(2^+) Verhandl. DPG (VI) 27 (1992) 188

M. Trautmannsheimer, P. Komninos, E. Huenges, H. Morinaga 63Ni-Mössbauer experiments at liquid helium temperature Verhandl. DPG (VI) 27 (1992) 130

V. Wagner, A. Kugler, M. Pacher, S. Hlavac, R. Wohlgemuth, W. Hering, R.S. Simon Response of BaF_2 Szintillator to High Energy Neutrons from Relativistic Heavy Ions Verhandl. DPG (VI) 27 (1992) 98

– 224 –

F. Weber and M.K. Weigel Impact of the Nuclear Equation of State on Models of General Relativistic Rotating Neutron Stars Verhandl. DPG (VI) 27 (1992) 230

A. Weigel, H. Lenske, H.H. Wolter Form factors for transfer reactions to the continuum Verhandl. DPG (VI) 27 (1992) 206

J. Werner, P. Ackerbauer, W.H. Breunlich, M. Fuchs, S. Fussy, M. Jeitler, P. Kammel, J. Marton, A. Scrinzi, J. Zmeskal, C. Petitjean, J. Bistirlich, K.M. Crowe, J. Kurck, R.H. Sherman, H. Bossy, H. Daniel, F.J. Hartmann, G. Schmidt, W. Neumann Experimental analysis of the steady state of muon-catalyzed d/t-fusion Verhandl. DPG (VI) 27 (1992) 31

M. Wieland, J. Fernández-Niello, F. Rieß, M. Aïche, A. Chevalier, J. Chevalier, J.C. Sens, N. Schulz, Ch. Briançon, E. Ruchowska, R. Kulessa Side bands in^{219}Ra Verhandl. DPG (VI) 27 (1992) 211

Th. Winkelmann, W. Assmann, K. Balog, P. Jänker, H. Leitz, K.E.G. Löbner, S.J. Skorka, H.-G. Thies Transfer Around the Coulomb-Barrier Verhandl. DPG (VI) 27 (1992) 106

H.-J. Wollersheim, P. Egelhof, H. Emling, J. Gerl, W. Henning, R. Holzmann, R. Schmidt, R.S. Simon, N. Martin, G. Eckert, Th.W. Elze, K. Stelzer, R. Kulessa, G. Duchene, B. Haas, J.C. Merdinger, J.P. Vivien, J. de Boer, E. Hauber, K. Kaiser, P. von Brentano, R. Reinhardt, R. Wirowski, R. Julin, C. Fahlander, I. Thorslund, H. Kluge Evidence for a Two-Phonon Octupole Vibrational State in^{208}Pb Verhandl. DPG (VI) 27 (1992) 210

T. Yamazaki, E. Widmann, R.S. Hayano, M. Iwasaki, S.N. Nakamura, K. Shigaki, H. Daniel, F.J. Hartmann, P. Hofmann, Y.S. Kim, T. von Egidy, J. Eades Exotic trapping of antiprotons in gaseous helium Verhandl. DPG (VI) 27 (1992) 33

Higher European Research Course for Users of Large Experimental Systems (HERCULES) Grenoble, France, February 24 - April 10, 1992

G. Schütz X-ray magnetic dichroism (S5) Proceedings

– 225 –

Rundgespräch ”Hochenergetische Spektroskopie elektronischer Zustände in Festkörpern und Molekülen” Bad Honnef, Germany, March 05 - 06, 1992

G. Schütz Untersuchung magnetischer Materie mit zirkularpolarisierter Röntgenstrahlung Proceedings

56. Physikertagung der DPG, AK Festkörperphysik (AG Magnetismus) Regensburg, Germany, March 16 - 20, 1992

G. Dollinger, P. Maier-Komor, Ch.M. Frey, E. Hammann Laserplasmaabscheidung dünner Kohlenstoffschichten Verhandl. DPG (VI) 27 (1992) 390

G. Dollinger, T. Faestermann, P. Maier-Komor, M. Boulouednine, H.-J. Körner Tiefenprofilanalyse mit hochenergetischen schweren Ionen Verhandl. DPG (VI) 27 (1992) 409

P. Fischer, S. Scherle, G. Schütz, M. Knülle, K. Attenkofer, S. Stähler, G. Wiesinger Magnetische XANES von Ho-Metall, Ho_3Fe_5O_{12}, Ho_2Co_{17} und Ho_6Fe_{23} Verhandl. DPG (VI) 27 (1992) 666

G. Schütz Bestimmung lokaler magnetischer Momente in Co/Fe-Oxiden Verhandl. DPG (VI) 27 (1992) 666

G. Schütz, K. Attenkofer, P. Fischer, M. Knülle, S. Stähler, B. Scholz, R.A. Brand, W. Keune Lokale magnetische Momente in TbFe Dünnschichtsystemen Verhandl. DPG (VI) 27 (1992) 688

S. Stähler, H. Ebert, P. Fischer, M. Knülle, G. Schütz Messung orbitaler Beiträge zum magnetischen Moment in Übergangsmetallen Verhandl. DPG (VI) 27 (1992) 666

International Workshop on Dynamical Fluctuations and Correlations in Nuclear Collisions Aussois, France, March 16-20, 1992

C. Fuchs, L. Sehn, H.H. Wolter Relativistic Kinetic Calculations of Heavy Ion Collisions with Phase Space Dependent Fields Nucl. Phys. A545 (1992) 151c

– 226 –

Frühjahrstagung Atomphysik, Kurzzeitphysik, Massenspektrometrie, Molekülphysik, Plasmaphysik, Quantenoptik der DPG Hannover, Germany, March 23 - 27, 1992

B. Busch, A. Ulrich, J. Wieser Verstärkte Spontanemission des 2. Excimerkontinuums von Xenon bei Ionenstrahlanregung Verhandl. DPG (VI) 27 (1992) 1424

J. Wieser, A. Ulrich, B. Busch, W. Krötz, G. Ribitzki Optische Verstärkung auf der 476,5 nm Ar II Laserlinie in einem schwerionenstrahlgepumpten Gastarget Verhandl. DPG (VI) 27 (1992) 1423

3rd European Particle Accelerator Conference Berlin, March 24 -28, 1992

A. Facco, K. Rudolph, A. Battistella, G. Tombola, F. Scarpa, A. Zanon The ALPI Linac Pulsing System Proceedings, Vol. 2

E. Nolte, G. Korschinek, W. Rühm, S. Neumaier, B. Dockhorn, T. Brunner, L. Zerle Accelerator Mass Spectrometry in Interdisciplinary Research Proceedings

56. Physikertagung der DPG Berlin, March 30 - April 3, 1992

J.D. Francke, W. Hering, R. Dengler Mechanisches Analog-Modell einer Hochfrequenzionenfalle (Paul-Falle) Verhandl. DPG (VI) 27 (1992) 1623

10th Intern. Conference on Plasma Surface Interactions in Controlled Fusion Devices Monterey, California, USA, March 30 - April 4, 1992

E. Hechtl, A. Mazanec, W. Eckstein, J. Roth, and C. Garcia-Rosales Sputtering behavior of boron and boron carbide J. Nucl. Mat. 196-98 (1992) 713

C.H. Wu, and E. Hechtl Erosion of tungsten by self-sputtering and light ion irradiation at oblique angles of incidence J. Nucl. Mat. 196-98 (1992) 713

– 227 –

Workshop on Muonic Atoms and Molecules Monte Veritá, Ascona, Switzerland, April 5 - 9, 1992

F.J. Hartmann The exotic-atom cascade Proceedings, p. 157, Birkhäuser Verlag Basel

K. Lou, C. Petitjean, T. Case, K.M. Crowe, W.H. Breunlich, M. Jeitler, P. Kammel, B. Lauss, J. Marton, W. Prymas, J. Zmeskal, D.V. Balin, V.N. Baturin, Yu.S. Grigoriev, A.I. Ilyin, E.M. Maev, G.E. Petrov, G.G. Semenchuk, Yu.V. Smirenin, A.A. Vorobyov, N.I. Voropaev, P. Baumann, H. Daniel, F.J. Hartmann, M. Mühlbauer, W. Schott, P. Wojciechowski Direct measurement of sticking in muon catalyzed dt fusion and physics of hot atoms Proceedings, p. 147, Birkhäuser Verlag Basel

M. Mühlbauer, H. Daniel, F.J. Hartmann A device for cooling charged particle beams by moderation plus acceleration Proceedings, p. 325, Birkhäuser Verlag Basel

P. Wojciechowski, P. Baumann, H. Daniel, F.J. Hartmann, M. Mühlbauer, W. Schott, A. Fuchs, P. Hauser, K. Lou, C. Petitjean, D. Taqqu, F. Kottmann Measurement of the stopping power for μ^− and μ^+ at energies between 3 keV and 50 keV Proceedings, p. 345, Birkhäuser Verlag Basel

EPS 12th General Conference of the Cond Matt Division Prague, CSFR, April 06 - 09, 1992

P. Fischer, G. Schütz, G. Wiesinger Magnetic Absorption at L_{2,3}-edges in Ho Proceedings

3rd International Colloquium on X-Ray Lasers Schliersee, Germany, May 18 - 22, 1992

A. Ulrich, J. Wieser, B. Busch, R. Gernhäuser, W. Krötz, G. Ribitzki, M. Salvermoser, D.E. Murnick Ion beam pumping of an ion laser line in a cold dense gas target Proceedings

– 228 –

Specialist Conference on Physics of Nuclear Induced Plasmas and Problems of Nuclear Pumped Lasers Obninsk, Russia, May 26 - 29, 1992

W. Krötz, A. Ulrich, B. Busch, G. Ribitzki, J. Wieser Third Excimer Continua in Neon and Argon Proceedings

A. Ulrich Lasers Pumped by Heavy-Ion Beams Proceedings

J. Wieser, A. Ulrich, B. Busch, W. Krötz, G. Ribitzki, D.E. Murnick Optical Gain on the 476.5 nm Argon-Ion Laser Line in Gas Target Excited by a Heavy-Ion Beam Proceedings

6th International Conference on Electrostatic Accelerators and Associated Boosters Montegrotto Terme, Padova, Italy, June 1 - 5, 1992

G. Dollinger, P. Maier-Komor, Ch.M. Frey, E. Hammann, H.-J. Körner New plant for fabricating carbon stripper foils by laser plasma ablation deposition Proceedings

T. Henkelmann, G. Korschinek Generation of dense Negative Ion Currents from a Laser Produced Plasma Proceedings

3rd International Conference on Application of Nuclear Techniques Mykonos, Greece, June 6 - 13, 1992

K. Knie, T. Faestermann, S. Kastel, D. Müller, G. Korschinek Detection of^{59}Ni by means of AMS: A study to determine the α particles in solar cosmic rays during the last 90 000 years Proceedings

E. Nolte Accelerator mass spectrometry with Heavy Ions: Global fall-out of cosmogenic^{36}Cl: The Pauli exclusion principle Proceedings

– 229 –

International Symposium on Weak and Electromagnetic Interactions in Nuclei (WEIN-92) Dubna, Moscow Region, Russia, June 16 - 22, 1992

H. Daniel Muon catalyzed fusion and energy production Proceedings

International Summer School on Nuclear Astrophysics Tianjin, P.R. of China, June 17 - 27, 1992

F. Weber and N.K. Glendenning Hadronic Matter and Rotating Relativistic Neutron Stars Proceedings, to be publ. by World Scientific, Singapore

3rd International School on Nuclear Physics ”New Trends in Nuclear Physics” Kiev, GUS, June 22 - July 1, 1992

H.H. Wolter Statistical Theory of Multistep Nuclear Reactions Proceedings, to be publ.

SIS-Theoretiker Treffen V Rauischholzhausen, Germany, June 26 - 28, 1992

C. Fuchs Relativistische LV Rechnungen mit zustandsabhängigen Feldern

R. Fauser Nichtgleichgewichts-Transporttheorie

M. Haft Relativistische direkte Schwerionenreaktionen

– 230 –

International Workshop on Muon Catalyzed Fusion μCF-92 Uppsala, Sweden, June 28 - July 3, 1992

W.H. Breunlich, P. Ackerbauer, M. Fuchs, S. Fussy, M. Jeitler, P. Kammel, B. Lauss, J. Marton, W. Prymas, J. Werner, J. Zmeskal, K. Lou, C. Petitjean, P. Baumann, H. Daniel, F.J. Hartmann, W. Schott, T. von Egidy, P. Wojciechowski, E.D. Bovet, D. Chatellard, J.-P. Egger, E. Jeannet, T. Case, K.M. Crowe, R.H. Sherman, V. Markushin Survey of experimental results on μCF including hyperfine effects Book of Abstracts, p. 10, ed. P. Froelich

H. Daniel Muon Catalyzed Fusion: Old and new aspects of energy production Book of Abstracts, p. 19, ed. P. Froelich

F.J. Hartmann, P. Ackerbauer, P. Baumann, W.H. Breunlich, H. Bossy, K.M. Crowe, H. Daniel, T. von Egidy, M. Fuchs, S. Grunewald, M. Jeitler, P. Kammel, R. Lipowsky, K. Lou, V. Markushin, J. Marton, E. Moser, N. Nägele, C. Petitjean, W. Schott, R.H. Sherman, J. Werner, J. Zmeskal The PSI experiments on muon-catalyzed pt fusion Book of Abstracts, p. 25, ed. P. Froelich

M. Jeitler, W.H. Breunlich, M. Cargnelli, P. Kammel, J. Marton, P. Pawlek, J. Werner, J. Zmeskal, K.M. Crowe, J. Kurck, A. Janett, C. Petitjean, R.H. Sherman, H. Bossy, H. Daniel, W. Neumann, M.P. Faifman, V.S. Melezhik, A. Adamczak Epithermal effects in muon-catalyzed dt fusion Book of Abstracts, p. 32, ed. P. Froelich

P. Kammel, P. Ackerbauer, J. Werner, W.H. Breunlich, M. Cargnelli, S. Fussy, M. Jeitler, J. Marton, A. Scrinzi, J. Zmeskal, T. Case, K.M. Crowe, K. Lou, C. Petitjean, R.H. Sherman, P. Baumann, H. Bossy, H. Daniel, F.J. Hartmann, W. Schott, T. von Egidy, W. Neumann The kinetics of muon-catalyzed dt fusion Book of Abstracts, p. 39, ed. P. Froelich

K. Lou, D.V. Balin, V.N. Baturin, P. Baumann, W.H. Breunlich, T. Case, K.M. Crowe, H. Daniel, Yu.S. Grigoriev, F.J. Hartmann, A.I. Ilyin, M. Jeitler, P. Kammel, B. Lauss, E.M. Maev, J. Marton, M. Mühlbauer, C. Petitjean, G.E. Petrov, W. Prymas, W. Schott, G.G. Semenchuk, Yu.V. Smirenin, A.A. Vorobyov, P. Wojciechowski, J. Zmeskal Final dt sticking ω_s using the ”survived muon method” and other new results from H/D/T triple mixtures Book of Abstracts, p. 49, ed. P. Froelich

M. Mühlbauer, H. Daniel, F.J. Hartmann Beam Cooling by Frictional Forces: Monte Carlo Calculations and Design Studies Book of Abstracts, p. 60, ed. P. Froelich

– 231 –

C. Petitjean, D.V. Balin, V.N. Baturin, P. Baumann, W.H. Breunlich, T. Case, K.M. Crowe, H. Daniel, Yu.S. Grigoriev, F.J. Hartmann, A.I. Ilyin, M. Jeitler, P. Kammel, B. Lauss, K. Lou, J. Marton, M. Mühlbauer, G.E. Petrov, W. Prymas, W. Schott, G.G. Semenchuk, Yu.V. Smirenin, A.A. Vorobyov, P. Wojciechowski, J. Zmeskal Experimental survey of the sticking problem in muon catalyzed dt fusion Book of Abstracts, p. 65, ed. P. Froelich

W. Schott, V.N. Baturin, W.H. Breunlich, T. Case, H. Daniel, F.J. Hartmann, M. Jeitler, P. Kammel, K. Lou, E.M. Maev, J. Marton, M. Mühlbauer, C. Petitjean, G.E. Petrov, Yu.V. Petrov, G.G. Semenchuk, Yu.V. Smirenin, A.A. Vorobyov, P. Wojciechowski, J. Zmeskal Feasibility of an experiment to determine the branching ratio for the emission of a heavy neutrino after muon capture in^3He Book of Abstracts, p. 73, ed. P. Froelich

P. Wojciechowski, P. Baumann, H. Daniel, F.J. Hartmann, C. Herrmann, M. Mühlbauer, W. Schott, A. Fuchs, P. Hauser, K. Lou, C. Petitjean, D. Taqqu, F. Kottmann Measurement of the stopping power for μ^− and μ^+ at energies between 3 keV and 50 keV Book of Abstracts, p. 82, ed. P. Froelich

Nobel Symposium No. 85 Saltsjöaden, Sweden, June 29 - July 3, 1992

P. Kienle Latest Hot Topics in Atomic Physics Research at GSI Proceedings

International Conference on Nuclei in the Cosmos Karlsruhe, Germany, July 6 - 10, 1992

F. Weber and N.K. Glendenning Interpretation of Rapidly Rotating Pulsars Proceedings, to be published by The Journal of Physics (England), (LBL-32562)

F.Weber and N.K. Glendenning Rapid Rotation of Pulsars

– 232 –

XIII International Conference on Cyclotrons and their Applications Vancouver, Canada, July 6 - 10, 1992

U. Trinks The Superconducting Separated-Orbit Cyclotron Tritron Proceedings

Gordon Conference on Electron Spectroscopy Brewster Academy, Wolfeboro, NH, USA, July 13 - 17, 1992

G. Schütz Magnetic X-ray Circular Dichroism Proceedings

6th International Conference on Nuclei far from Stability and 9th International Conference on Atomic Masses and Fundamental Constants Bernkastel-Kues, Germany, July 19 - 24, 1992

E. Hagn Measurements of Nuclear Moments with NMR-ON Book of Abstracts

F. Heine, T. Faestermann, A. Gillitzer, H.-J. Körner Proton and Alpha Radioactivity in the Vicinity of^{100}Sn Proceedings

G. Kraus, P. Egelhof, C. Fischer, H. Geissel, A. Himmler, F. Nickel, G. Münzenberg, W. Schwab, A. Weiß, J. Friese, A. Gillitzer, H.-J. Körner, M. Peter, W. Henning, J.P. Schiffer, J.V. Kratz, L.V. Chulkov, M. Golovkov, A.A. Ogloblin Investigation of proton inelastic scattering on^{56}Ni in inverse kinematics Proceedings

H. Lenske Dynamical Correlations in Drip Line Nuclei Proceedings

J. Trötscher, H. Wollnik, K. Balog, H. Eickhoff, B. Franczak, B. Franzke, Y. Fujita, H. Geissel, Ch. Klein, K.E.G. Löbner, A. Magel, G. Münzenberg, A. Przewloka, T. Radon, D. Rosenauer, H. Schäfer, M. Sendor, D.J. Vieira, B. Vogel, Th. Winkelmann Developments for Direct Mass Measurement of Exotic Nuclei at the ESR Book of Abstracts, PH15

– 233 –

International Conference on the Physics of Transition Metals, ICPTM 92 Techn. Univ. Darmstadt, Germany, July 20 - 24, 1992

G. Schütz, P. Fischer, S. Stähler, M. Knülle Experimental Studies of Circular Magnetic X-ray Dichroism Proceedings

International Nuclear Physics Conference Wiesbaden, Germany, July 26 - August 1, 1992

K. Balog, B. Franczak, B. Franzke, Y. Fujita, H. Geissel, Ch. Klein, J. Knollmann, K.E.G. Löbner, A. Magel, G. Münzenberg, A. Przewloka, D. Rosenauer, H. Schäfer, H. Seifert, M. Sendor, J. Trötscher, D.J. Vieira, B. Vogel, Th. Winkelmann, H. Wollnik Planned Multi-Turn Time-of-Flight Mass Measurements of Exotic Nuclei in the Experimental Storage Ring of the GSI Book of Abstracts, 5.28

K. Balog, B. Franczak, B. Franzke, Y. Fujita, H. Geissel, Ch. Klein, J. Knollmann, K.E.G. Löbner, A. Magel, G. Münzenberg, A. Przewloka, D. Rosenauer, H. Schäfer, H. Seifert, M. Sendor, J. Trätscher, D.J. Vieira, B. Vogel, Th. Winkelmann, H. Wollnik Development of a Fast Time-Recording- and Data Acquisition System for Direct Mass Measurements of Exotic Nuclei at the ESR Book of Abstracts, 5.29

K. Balog, P. Jänker, H. Leitz, U. Lenz, K.E.G. Löbner, S.J. Skorka, H.-G. Thies, Th. Winkelmann, L. Corradi, D.R. Napoli, A.M. Stefanini, S. Beghini, G. Montagnoli, F. Scaralassara, C. Signorini, F. Soramel Fusion and Transfer at Near-Barrier Energies Book of Abstracts, 3.2.55

T. von Egidy, P. Baumann, H. Daniel, T. Haninger, F.J. Hartmann, P. Hofmann, Y.S. Kim, M.S. Lotfranaei, W. Schmid, A.S. Botvina, Ye. Golubeva, A.S. Iljinov, V.G. Nedorezov, A.S. Sudov, J. Jastrzebski, W. Kurcewicz, P. Lubinski, A. Grabowska, A. Stolarz, D. Hilscher, D. Polster, H. Rossner, H. Machner, G. Riepe, P. David, H.S. Plendl, J. Lieb, B. Wright, K. Ziock Nuclear heating and fission induced by antiproton annihilation Book of Abstracts, 3.2.44

J. Friese, H.-J. Körner, J. Reinhold, R. Schneider, H. Trieb, K. Zeitelhack, B. Blank, T. Brohm, Y. Fujita, H. Geissel, W. Koenig, G. Münzenberg, F. Nickel, M. Pfützner, K. Rykaczewski, I. Schall, D. Schardt, A. Schröter, M. Steiner, K. Sümmerer, B. Voss, J. Weckenmann Production of Neutron Rich Nuclei by Fragmentation of 760 A·MeV^{136}Xe Proceedings

– 234 –

C. Fuchs, L. Sehn, H.H. Wolter Relativistic kinetic calculations with momentum configuration dependent fields Book of Abstracts, 3.4.28

T. Haninger, F.J. Hartmann, P. Hofmann, Y.S. Kim, M.S. Lotfranaei, T. von Egidy, H. Märten, A. Ruben Folding angle of fragments and excitation energy in low energy fission Book of Abstracts, 3.2.37

R.R. Hilton, W. Höhenberger, and P. Ring Nature of K=1^+-States in Rare Earth Nuclei Book of Abstracts, 1.1.34

P. Jänker, K. Balog, H. Leitz, U. Lenz, K.E.G. Löbner, M. Moralles, H.-G. Thies, Th. Winkelmann Reduction of the Evaporation Barriers in the Compound Nuclei Decay Book of Abstracts, 3.2.52

H. Lenske, R. Siudak, H.H. Wolter Statistical Direct Reaction Theory for Preequilibrium and Dissipative Collisions Book of Abstracts, 3.3.24

H. Lenske, M. Haft, H.H. Wolter Covariant Direct Reaction Theory for Exclusive Heavy Ion Collisions Book of Abstracts, 3.4.26

M. Reboiro and P. Ring Mapping of Multi-Quark Configurations to Baryons and Mesons Book of Abstracts, 2.1.1

A. Schröter, E. Berdermann, H. Geissel, A. Gillitzer, J. Homolka, P. Kienle, W. Koenig, B. Povh, F. Schumacher, H. Ströher Subthreshold Antiproton Production in Heavy Ion Collisions at SIS-Energies Book of Abstracts, 3.5.16

M.M. Sharma and P. Ring Neutron Skin in Relativistic and Non-Nucleonic Mean Field Theory Book of Abstracts, 1.4.32

M.M. Sharma and P. Ring Relativistic Mean-Field Description fo Neutron-Deficient Pt Isotopes Method Book of Abstracts, 1.6.11

Y. Sun, K. Hara, and P. Ring Removal of Spurious States in the Projected Tamm-Dancoff Method Book of Abstracts, 1.6.19

– 235 –

XXVI International Conference on High Energy Physics Dallas, USA, August 5 - 12, 1992

D.B. Ion and W. Stocker Coherent Photon and Meson Production via Cherenkov like Effects in Heavy Ion Interactions Contributed paper n.230

International Conference on Anomalous Scattering Malente, Hamburg, Germany, August 17 - 21, 1992

G. Schütz, P. Fischer, S. Stähler, M. Knülle, R. Frahm, H. Ebert Magnetic Absorption of Circularly Polarized Hard X-rays Proceedings

G. Schütz, P. Fischer, S. Stähler, M. Knülle, R. Frahm, H. Ebert High Precision Measurement of Circular Magnetic Dichroism Proceedings

7th International Conference on X-ray Absorption Fine Structure Kobe, Japan, August 23 - 29, 1992

M. Knülle, G. Schütz, S. Parkin, P. Fischer, S. Stähler, H. Ebert Distribution of Magnetic Moments in Co/Pt and Co/Pt/Ir/Pt Multilayers Detected by Magnetic Absorption Proceedings

M. Knülle, R. Frahm, S. Stähler, P. Fischer, G. Schütz SPEXAFS at the L_{2/3}-edges of Eu_3Fe_5O_{12}, Gd- and Tb-metal and the Alloy HoGd Proceedings

G. Schütz Circular Magnetic X-ray Dichroism: New Aspects Proceedings

– 236 –

2nd International Conference on Particle Production near Threshold Uppsala, Sweden, August 26 - 29, 1992

A. Schröter, E. Berdermann, A. Gillitzer, J. Homolka, P. Kienle, W. Koenig, B. Povh, F. Schumacher, H. Ströher Subtreshold Antiproton Production in Heavy Ion Collisions Proceedings, to be published in Physica Scripta

NATO Advanced Study Institute on ”Topics in Atomic and Nuclear Collisions” Predeal, Romania, August 31 - September 11, 1992

W. Stocker and D.B. Ion Electromagnetic and Mesonic Cherenkov Mechanisms in Nuclear Media Proceedings, invited lecture

C. Fuchs Covariant kinetic calculations of heavy ion collisions Proceedings

Symposium on Magnetic Ultra Thin Films, Multilayers and Surfaces Lyon, France, September 7 - 10, 1992

G. Schütz, S. Stähler, P. Fischer, S. Parkin, M. Knülle, H. Ebert Distribution of Magnetic Moments in Co/Pt and Co/Pt/Ir multilayers detected by magnetic absorption (BPIV-14) Proceedings

International School of Nuclear Physics on ”Heavy Ion Collisions at Intermediate and Relativistic Energies” Erice, Sicily, Italy, September 7 - 16, 1992

A. Gillitzer, A. Schröter, E. Berdermann, J. Homolka, P. Kienle, W. Koenig, B. Povh, F. Schumacher, H. Ströher Subthreshold Antiproton Production in Heavy Ion Collisions Proceedings

H.H. Wolter Relativistic Kinetic Calculations using Phase Space dependent Fields Proceedings, Progress in Particle and Nuclear Physics, Vol.30

– 237 –

European Research Conference on Electronic Structure of Solids Cambridge, England, September 13 - 17, 1992

G. Schütz Magnetic Studies with Circularly Polarized X-rays Proceedings

2nd Biennial Conference on Low-Energy Antiproton Physics, LEAP’92 Courmayeur, France, September 14 - 19, 1992

T. von Egidy, H. Daniel, F.J. Hartmann, P. Hofmann, Y.S. Kim, W. Schmid, H.H. Schmidt, G. Riepe, H. Machner, P. David, H.S. Plendl, A.S. Iljinov, A.S. Botvina, Ye. Golubeva, M.V. Mebel, D. Hilscher, D. Polster, H. Rossner, J. Jastrzebski, W. Kurcewicz Antiproton induced fission Proceedings, to be published in Nucl. Phys. A

J. Jastrzebski, H. Daniel, T. von Egidy, A. Grabowska, Y.S. Kim, W. Kurcewicz, P. Lubinski, G. Riepe, W. Schmid, A. Stolarz, S. Wycech Signature of a neutron halo in^{232}Th from antiproton absorption Proceedings, to be published in Nucl. Phys. A

E. Widmann, I. Sugai, T. Yamazaki, R.S. Hayano, M. Iwasaki, S.N. Nakamura, H. Tamura, T. Ito, A. Kawachi, N. Nishida, W. Higemoto, Y. Ito, N. Morita, F.J. Hartmann, H. Daniel, T. von Egidy, W. Schmid, J. Hoffmann, and J. Eades Antiproton trapping in various helium media: report of the HELIUMTRAP experiment at LEAR Proceedings, to be published in Nucl. Phys. A

International Conference on Nuclear Structure and Reactions at Low and Intermediate Energies Dubna, GUS, September 15 - 19, 1992

R.R. Hilton Is there an Angular Analogue to the Giant Dipole Resonance? Proceedings, invited talk

H. Hofmann Collective motion as a transport process: the implications of finite temperature Proceedings, invited talk

– 238 –

H. Lenske Nuclear Structure and reactions of exotic nuclei Proceedings, to be publ.

P. Ring Static and Dynamic Relativistic Description of the Structure of Finite Nuclei Proceedings

Arbeitstreffen Mittelenergiephysik Manderscheid, Eifel, Germany, September 21 - 25, 1992

T. von Egidy, H. Daniel, F.J. Hartmann, P. Hofmann, Y.S. Kim, W. Schmid, H. Machner, G. Riepe, P. David, D. Hilscher, D. Polster, H. Rossner, H.S. Plendl, J. Jastrzebski, W. Kurcewicz, A.S. Iljinov, A.S. Botvina, M.V. Mebel, Ye. Golubeva Heiße Kerne und Spaltung durch Antiproton-Annihilation Proceedings, p. 377

F.J. Hartmann, T. Yamazaki, E. Widmann, R.S. Hayano, M. Iwasaki, S.N. Nakamura, K. Shigaki, H. Daniel, T. von Egidy, P. Hofmann, Y.S. Kim, W. Schmid, J. Eades Langlebige Zustände in antiprotonischem Helium Proceedings, p. 371

F.J. Hartmann, D.V. Balin, V.N. Baturin, P. Baumann, W.H. Breunlich, T. Case, K.M. Crowe, H. Daniel, Yu.S. Grigoriev, A.I. Ilyin, M. Jeitler, P. Kammel, B. Lauss, K. Lou, J. Marton, M. Mühlbauer, C. Petitjean, G.E. Petrov, W. Prymas, W. Schott, G.G. Semenchuk, Yu.V. Smirenin, A.A. Vorobyov, P. Wojciechowski, J. Zmeskal Die Haftwahrscheinlichkeit nach myonenkatalysierter dt-Fusion: Messungen mit einer Ionisationskammer Proceedings, p. 537

16th World Conference of the International Nuclear Target Development Society Padova, Italy, September 21 - 25, 1992

G. Dollinger, M. Boulouednine, T. Faestermann, P. Maier-Komor Depth microscopy for thin film analysis Proceedings

G. Dollinger, Ch.M. Frey, P. Maier-Komor Laser plasma ablation of carbon Proceedings

– 239 –

R. Grossmann and H.J. Maier Organization and radiation protection conceptions of the University of Munich hot-lab facility Proceedings

H.J. Maier, H.U. Friebel, D. Frischke, and R. Grossmann Preparation of self supporting, metallic Hf targets from isotopically enriched HfO_2 Proceedings

H.J. Maier, H.U. Friebel, D. Frischke, and R. Grossmann State of the art of high vacuum sputter deposition of nuclear targets Proceedings

P. Maier-Komor, K.-H. Speidel, A. Stolarz Targets of magnetic properties Proceedings

P. Robouch, P. Maier-Komor, J. Pauwels New development in target preparation at CBNM: laser evaporation-condensation and electrodeposition Proceedings

K.-H. Stephan, H. Bräuninger, C. Reppin, H.J. Maier, D. Frischke, M. Krumrey, and P. Möller Optical filter for X-ray astronomy CCDs Proceedings

Jahrestagung des Fachausschusses ”Festkörperphysik” der ÖPG Wien, Austria, September 22, 1992

G. Schütz Spinabhängige Absorption von Röntgenstrahlen - eine neue Methode zum Studium magnetischer Ordnung Proceedings

International Conference Tokyo, Japan, October 1992

Y.K. Gambhir, J.P. Maharana, J.A. Sheikh, and P.Ring Relativistic Mean Field Description of Exotic Nuclear Shapes and Deformations near Z=34

– 240 –

Workshop on ”The Production and Use of Intense Radioactive Ion Beams at the Isospin Laboratory” Oak Ridge, U.S.A, October 7 - 11, 1992

K.E.G. Löbner The ”Charge State Enforcer” Revisited Proceedings

IEEE 1992 Nuclear Sciences Symposium Orlando, FL, USA, October 25 - 31, 1992

M. Ciric, J. Friese, H.-J. Körner, M.R. Maier Low Cost Pulse Shaper for Use with Silicon PIN Diodes Proceedings

J. Friese, A. Gillitzer, H.-J. Körner, J. Reinhold, M. Peter, M.R. Maier The SISWICH, A Detector Telescope with Intrinsic Calibration Proceedings

COSY FFE Arbeitstreffen Jülich, Germany, October 27 - 28, 1992

W. Stocker Elektromagnetische und mesonische Cherenkov-Mechanismen in Proton-Kern-Kollisionen Proceedings

12th International Conference on the Application of Accelerators in Research and Industry Denton, Texas, USA, November 2 - 11, 1992

W. Assmann Target analysis with heavy ion beams Proceedings

G. Dollinger Elastic Recoil Detection Analysis with Atomic Depth Resolution Proceedings

H.J. Maier Modern aspects of nuclear accelerator target preparation Proceedings

– 241 –

P. Maier-Komor Carbon foils for nuclear accelerator experiments Proceedings

D. Müller Accelerator Mass Spectrometry With a Gasfilled Magnetic Spectrograph Proceedings

International Symposium on Negative Ion Sources Brookhaven, USA, November 9 - 13, 1992

G. Korschinek, T. Henkelmann The Munich Laser Ion Source for use of negative Ions Proceedings

E-MRS Fall Meeting Strasbourg, France, November 11 - 14, 1992

P. Berberich, W. Assmann, W. Prusseit, B. Utz, H. Kinder Large area deposition of YBaCuO films by thermal co-evaporation

Symposium: Beschleuniger-Massenspektrometrie und ihre Anwendungen University of Kiel, Germany, 30. November 1992

G. Korschinek Beschleuniger-Massenspektrometrie am Garchinger Tandem-Beschleuniger: Eine Übersicht zu den angewandten Methoden und Messungen

– 242 –

37th Annual Conference on Magnetism and Magnetic Materials Houston, TX, USA, December 1 - 4, 1992

K. Attenkofer, P. Fischer, G. Schütz, M. Knülle, S. Stähler, B. Scholz, R.A. Brand, W. Keune, U. Köbler, G. Wiesinger Mössbauer and X-ray Dichroism studies of the Magnetic Spin Structure in Fe/Tb multilayers Proceedings

P. Fischer, G. Schütz, W. Kaiser, J. Hunecke, M. Knülle, S. Stähler Magnetic Scattering of Circularly Polarized X-rays Proceedings

M. Knülle, P. Fischer, R. Frahm, G. Schütz, S. Stähler SPEXAFS at the L_{2/3}-edges of Eu_3Fe_5O_{12}, Gd- and Tb-metal and the Alloy HoGd Proceedings

G. Schütz, P. Fischer, S. Parkin, M. Knülle, S. Stähler, H. Ebert Distribution of Magnetic Moments in Co/Pt and Co/Pt/Ir multilayers detected by magnetic absorption Proceedings

S. Stähler, S. Welzel-Gerth, P. Fischer, M. Knülle, B. Buchholz, G. Schütz Temperature Dependence of the CMXD in the Invar Alloy Fe_{72}Pt_{28} Proceedings

International Conference to Celebrate Yakir Aharonov’s 60th Birthday, ”Fundamental Aspects of Quantum Theory” Columbia, S.C., USA, December 9 - 12, 1992

E. Nolte, M. Thoma, T. Faestermann, A. Gillitzer, G. Korschinek, D. Müller Search for small violations of the Pauli exclusion principle and the size of the electron

DAE Nuclear Physics Symposium Bompay, India, December 21 - 24, 1992

A. Ansari, R. Rossignoli, and P. Ring Spin Projected Partition Function and Nuclear Level